Author's response to reviews

Title: A qualitative study of changes in expectations over time among patients with chronic low back pain seeking four CAM therapies

Authors:

Emery R Eaves (emerye@email.arizona.edu)
Karen J Sherman (sherman.k@ghc.org)
Cheryl Ritenbaugh (ritenbau@email.arizona.edu)
Clarissa Hsu (hsu.c@ghc.org)
Mark Nichter (mnichter@email.arizona.edu)
Judith A Turner (jturner@u.washington.edu)
Daniel C Cherkin (cherkin.d@ghc.org)

Version: 2  Date: 15 August 2014

Author's response to reviews:

Dear Editor,

per your request, we have now added line numbers - there were page numbers already. lines 199 and 200 indicate that we have used pseudonyms.

best,

Karen Sherman